Author’s response to reviews

Title: Androgen secreting steroid cell tumor of the ovary in a young lactating women with acute onset of severe hyperandrogenism: a case report

Authors:

Altaf Gauhar (altafhaji786@yahoo.com)
Shekhar Sharma (drshekharsharma@gmail.com)
Manoj Babu (manojbabur@aims.amrita.edu)
Vijaykumar D K (dkvijaykumar@aims.amrita.edu)
K Chithrathara (chithrathara@aims.amrita.edu)

Version: 11 Date: 30 November 2007

Author’s response to reviews: see over
Dear Editor,

We have made the following changes as per your advice:

1. Author list –
   The list of authors in the manuscript are written exactly as they are in the submission system. The author’s details and arrangement are also corrected in the submission system.

2. Author email addresses
   Email addresses for each author are given on the title page, beneath the corresponding author details.

3. Abstract
   Abstract is structured into Introduction, Case presentation and Conclusions. The same has been updated on the submission page.

4. Authors' contributions
   The Authors contributions section is placed towards the end of the manuscript text, directly preceding the Acknowledgements. This section is edited as per the editor’s advise.

5. After the 'Conclusions' the sections are now in this order: Competing interests; Authors' contributions; Consent; Acknowledgements

6. References - The reference list contains all authors' names.

7. References are cited in consecutive numerical order.

Thanks,

Dr. Altaf Gauhar Haji